Author's response to reviews

Title: Kaplan anastomosis of the ulnar nerve. A case report

Authors:

Georgios GP Paraskevas (g_paraskevas@yahoo.gr)
Christos ChG Gekas (gekaschris@yahoo.gr)
Alexandros AT Tzaveas (tzaveas@hotmail.com)
Ioannis IS Spyridakis (jaspyr@otenet.gr)
Alexandra AS Stoltidou (alexmed501@yahoo.com)
Parmenion PPh Tsitsopoulos (par_tsis@yahoo.gr)
Orestis OI Ioannidis (telonakos@yahoo.gr)

Version: 2 Date: 7 September 2007

Author's response to reviews:

Dear Sirs,

Thank you for your letter. Accordingly we undertook the following changes:
- The background section has been renamed as introduction
- An acknowledgement as well as an consent section has been included

Awaiting for your review,

Sincerely,

George Paraskevas, M.D.